

JGR Solid Earth

RESEARCH ARTICLE

10.1029/2022JB026202

Key Points:

- The MLD may be caused by channel flow in the cratonic mantle lithosphere in the Archean when the mantle was hot
- The top of the channel flow can cause radial anisotropy and reduce seismic wave velocity
- Secular cooling of the earth can preserve the top of the shear zone caused by the channel flow

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

H. Yang, haibin.yang@anu.edu.au

Citation:

Yang, H., Artemieva, I. M., & Thybo, H. (2023). The mid-lithospheric discontinuity caused by channel flow in proto-cratonic mantle. *Journal of Geophysical Research: Solid Earth*, *128*, e20221B026202. https://doi. org/10.1029/2022JB026202

Received 5 DEC 2022 Accepted 14 MAR 2023

Author Contributions:

Conceptualization: Haibin Yang Investigation: Haibin Yang Methodology: Haibin Yang Supervision: Irina M. Artemieva, Hans Thybo Visualization: Haibin Yang, Irina M.

Artemieva Writing – original draft: Haibin Yang

Writing – review & editing: Irina M. Artemieva, Hans Thybo

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

caused by channel c mantle lithosphere en the mantle was hot or lithow a hot

in Proto-Cratonic Mantle

University, Canberra, ACT, Australia, ³SinoProbe Laboratory, Chinese Academy of Geological Sciences, Beijing, China, ⁴Section of Marine Geodynamics, GEOMAR Helmholtz Center for Ocean Research, Kiel, Germany, ⁵State Key Laboratory of Geological Processes and Mineral Resources, School of Earth Sciences, China University of Geosciences, Wuhan, China, ⁶Eurasia Institute of Earth Science, Istanbul Technical University, Istanbul, Turkey

The Mid-Lithospheric Discontinuity Caused by Channel Flow

Abstract Global geophysical observations show the presence of the enigmatic mid-lithospheric discontinuity (MLD) at depths of ca. 80–150 km which may question the stability and internal structure of the continental lithosphere. While various mechanisms may explain the MLD, the dynamic processes leading to the seismic observations are unclear. Here we present a physical mechanism for the origin of MLD by channel flow in the cratonic mantle lithosphere, triggered by convective instabilities at cratonic margins in the Archean when the mantle was hot. Our numerical modeling shows that the top of the frozen-in channel flow creates a shear zone at a depth comparable to the globally observed seismic MLD. Grain size reduction in the shear zone and accumulation of percolated melts or fluids along the channel top may reduce seismic wave speeds as observed below the MLD, while the channel flow itself may explain radial anisotropy of seismic wave speeds and change in direction of the seismic anisotropic fast axis. The proposed mechanism is valid for a broad range of physically realistic parameters and that MLD may have been preserved since its formation in the Archean. The intensity of the channel flow ceased with time due to secular cooling of the Earth's interior. The new mechanism may reshape our understanding of the evolution and stability of cratonic lithosphere.

Plain Language Summary Stable cratons form rheologically strong plates atop a ductile asthenospheric mantle. Various types of seismic observations show the presence of a potentially rheologically weak zone at depths of ca. 80–150 km termed the mid-lithospheric discontinuity (MLD). While several mechanisms may explain the present-day MLD seismic structure, the dynamic processes that created such structure are unclear. We propose that the MLD was created by channel flow in the lower cratonic lithosphere, triggered by convective instabilities at cratonic margins in the Archean when the mantle was hot. The presence of a chemically distinct, low-density cratonic lithospheric root is required to initiate the process. We test our proposed model by numerical modeling and demonstrate that the proposed mechanism is valid (a) for a broad range of physically possible parameters in nature and (b) that MLD may have been preserved since its formation in the Archean.

1. Introduction

The depth to the lithosphere-asthenosphere boundary (LAB) in stable cratons is debated and is typically interpreted at depths between ca. 200 and 400 km, as estimated from heat flow analysis (Artemieva & Mooney, 2001), mantle xenoliths (O'Reilly et al., 2001), seismological data (Gung et al., 2003), magnetotelluric data (Jones et al., 2003), and isostatically based evaluations (Koptev & Ershov, 2010). However, high-resolution seismological surveys have revealed more detailed and complex structures in the mechanically strong layer (Figure 1, discussion in Section 4).

Extensive high-density, long-range seismic sections detected intra-lithosphere discontinuities at depths around 90–130 km in the Baltic Shield (Perchuc & Thybo, 1996) and globally under stable continents (Thybo & Perchuć, 1997; Figure 1a). These seismic discontinuities mark the top of a seismic low-velocity zone (LVZ) with a small reduction in *P* wave velocity (Vp) and a larger reduction, of up-to 6%, in shear-wave velocity (Vs) over a depth range of 20–50 km in the Baltic Shield (Abramovitz et al., 2002). Similarly, the presence of one or multiple seismic LVZ with the top typically at depths of ca. 80–100 km and a thickness of 20–40 km is reported for the Siberian Craton based on the Soviet PNE long-range seismic refraction profiles (Egorkin et al., 1987). Beneath

Thickness of intra-lithosphere anomalous layers



Journal of Geophysical Research: Solid Earth





(b)

Figure 1. Compilation of depth to intra-lithosphere discontinuities in cratons (a) and thickness of the anomalous intra-lithosphere layer below these discontinuities (b) from seismic, MT, and xenolith data (see references in Supporting Information S1).

the North American craton, joint inversion of the high-frequency *P* wave coda with long period surface wave dispersion data identified not only a velocity discontinuity with a 2.1% Vs decrease at depths of ca. 100–120 km beneath the cratonic stations, but also a positive intra-lithospheric discontinuity at around 150–170 km depth with a 3.6% Vs increase (Calo et al., 2016).

The intra-lithosphere seismic discontinuity with a negative velocity converter at depths ca. 70–120 km globally under stable continents has also been identified by high-frequency seismic receiver function (RF) studies (Fischer et al., 2010; Rychert & Shearer, 2009). Similar studies, largely based on *S*-RF (*S* wave RF), reported a negative converter at ca. 60–85 km depth beneath the Precambrian Western Australia (Sun et al., 2018) and typically at ca. 80–120 km depth beneath the cratonic North America (Chen et al., 2018). Beneath the Kalahari craton, some *S*-RF studies identified a shallow (80–90 km depth) negative converter found in other *S*-RF studies (Rychert & Shearer, 2009), while other *S*-RF studies found a prominent interface with a 4.5% reduction in seismic velocity at ca. 150–160 km depth (Savage & Silver, 2008). Notably, the deep converter corresponds to the top of melt-metasomatized mantle beneath the Kaapvaal craton (Griffin et al., 2003; Figure 1b).

Vertical stratification of the cratonic lithosphere with the transition between compositionally distinct layers at ca. 120–150 km depth is revealed by mantle xenolith studies in the Archean-Paleoproterozoic provinces of the Slave craton and the Baltic Shield (Kopylova & Caro, 2004). Vertical stratification of the cratonic lithospheric mantle is also imaged in seismic azimuthal anisotropy studies of the North American and the Australian cratons, which show a boundary between mantle layers with different fast directions at ca. 100–150 km depth (Sun et al., 2018; Yuan & Romanowicz, 2010). Likewise, strong electrical anisotropy has been reported for the Superior Province with the top of the anisotropic layer at ca. 70–100 km depth (Mareschal et al., 1995), while in the Slave craton a pronounced decrease in electrical conductivity by 0.5 log unit at 135–150 km depth (Jones et al., 2003) closely correlates with the compositional boundary established from mantle xenoliths (Kopylova & Russell, 2000).

This brief overview demonstrates the presence of an intra-lithosphere heterogeneity at a depth of around 100 km in stable continental lithosphere globally, although with regional depth variations (Figure 1a). It is, however, still debated if the top of an intra-lithospheric seismic refraction LVZ and negative seismic convertors imaged by RFs (later referred to as the mid-lithospheric discontinuity, MLD) image the same intra-lithospheric transition zone, and if anisotropic layering and metasomatic imprint on composition of the cratonic lithospheric mantle correspond to the same feature.

Various models have been brought forward to explain the MLD. They include: (a) rheological layering at temperature close to the solidus in the presence of melt/fluids (Thybo, 2006), (b) compositional layering due

to either partial melt extraction (Yuan & Romanowicz, 2010), or basaltic melt additions and metasomatism (Artemieva, 2009; Rader et al., 2015; Thybo, 2006), (c) change in deformation style, either depth-dependent (Karato et al., 2015), or time-dependent such as transition from frozen-in (e.g., associated with mineral fabric inherited from cratonization) to young seismic anisotropy (Yuan & Romanowicz, 2010).

These mechanisms are under discussion. First, it is debated if solidus temperature may be reached at mid-lithosphere depths below stable continents even for a fertile and volatile-rich composition (Karato et al., 2015). Second, compositional layering may not necessarily relate to the seismic MLD. Minerals rich in carbon or amphibole may produce a >5% seismic wave velocity drop (Selway et al., 2015), but mantle xenoliths suggest that such minerals may not occur ubiquitously in cratonic lithosphere (Rader et al., 2015; Selway et al., 2015). Third, an analysis of feasible mechanisms for a seismic velocity change in *S*-RF suggests that radial anisotropy could be a strong candidate to explain the seismic MLD (Selway et al., 2015). However, this model requires a global operation of a geological process which produces the same anisotropic layering pattern at the same depth range in every craton, and such process is yet unknown. Azimuthal anisotropy may either increase or decrease seismic velocity depending on the back azimuth of seismic waves, but this is not revealed by long-range controlled-source seismic profiles in the cratons of North America, Siberia, and Baltica (Thybo, 2006). Elastically accommodated grain-boundary sliding and mantle hydration, for example, by melt metasomatism, seem to predict a velocity drop similar to observed at the MLD (Karato et al., 2015). However, the model indicates no relationship between a reduction in seismic velocity at the MLD and anisotropy variations within the lithospheric mantle.

In summary, mechanisms for the presence of the MLD should explain the following features: (a) a seismic velocity interface observed both in seismic controlled-source and RF models, (b) a transition between layers with different directions of seismic, and possibly electrical, anisotropy, and (c) the ubiquitous appearance of the MLDs in cratons.

Here, we propose a new mechanism for the origin of the MLD by frozen-in channel flow formed at the early cratonic evolution. This mechanism may resolve the controversy between various MLD models. We use a 2D state-of-the-art geodynamic modeling software "Underworld" (Moresi et al., 2007) to demonstrate that channel flow may explain the effects observed in various geophysical data and interpreted as the MLD. The numerical model predicts the formation of a rheologically weak mid-lithosphere zone in the cratons globally, with the associated changes in seismic velocity and formation of anisotropic layering. We propose that a global mid-lithosphere weakness zone may have formed in the Precambrian at the early stages of cratonization by channel flow in the mantle. We argue that the intensity of the channel flow mechanism ceased with time due to secular cooling of the Earth's interior.

2. Setup of Numerical Experiments

2.1. Thickness of Chemical and Thermal Boundary Layers

We perform numerical experiments (see Methods in Supporting Information S1) to examine the long-term evolution of Archean lithosphere for a range of possible thicknesses of the proto-lithospheric mantle thickness (Figure 2a and details below). Edge-driven convection around the transition between thick and thin lithosphere triggers a downwelling mantle flow, which creates a horizontal pressure gradient in the lower part of cratonic lithosphere. This, in turn, causes a Poiseuille-type channel flow which produces features similar to the MLD.

The model time zero corresponds to the Archean formation of a compositionally low-density continental lithospheric root below the craton, which forms the chemical boundary layer (CBL) composed of a buoyant material (Figure 2b), left behind after an extensive melt extraction from the mantle (Carlson et al., 2005). The base of the CBL defines the chemical LAB (CLAB). Besides, the lithosphere of both the thick and thin proto-CBL domains makes the conductive thermal boundary layer (TBL) atop the convecting mantle, and the boundary between the conductive and convective mantle defines the thermal LAB (TLAB).

Initially, CLAB is deeper than TLAB (Figure 2c). This situation corresponds to the start of cratonic evolution in the Archean (time zero), when a high concentration of radioactive elements in the chemical lithosphere of cratons and a high Archean geothermal gradient (Herzberg et al., 2010) cause partial melting of lower lithospheric mantle above the CLAB, so that the base of the conductive layer TLAB in the hot Earth is shallower than CLAB (Figure 2c). This creates a gravitationally unstable situation with a thick depleted, low-density, and at the same





Figure 2. Initial setup of the end-member model (Table S1 in Supporting Information S1), which includes a thick proto-lithosphere (x = 0-2,400 km) and a thin proto-lithosphere (x = 2,600-4,000 km) area with a linear transition between the two domains. (a) The model includes three layers: crust—top 40 km; mantle chemical boundary layer (CBL) with the base (chemical LAB, CLAB) at a depth of 235 km in the thick domain and 95 km in the thin domain, underlain by asthenospheric mantle down to 660 km depth. Circular markers in the upper half of the model (at x = 500-3,000 km) trace mantle deformation. (b) Initial in situ density profiles at thick (x = 2,300 km, blue) and thin proto-lithosphere domain (x = 2,800 km, red). In the sublithospheric mantle, a temperature-controlled density decrease with depth leads to convective instability (gray arrows). The lithostatic pressure difference between the two profiles is illustrated by the orange line. (c) Initial temperature profile at time zero with a linear temperature increase from 0°C at the model top to a potential temperature of 1,500°C at depth 90 km (thermal LAB, TLAB), with further increase in deeper mantle with an adiabatic gradient of 0.3° C/km.

time warm cratonic root above undepleted convective mantle (Figure 2b). The situation may have analogy with modern oceanic plateau when intensive mantle melting in regions with hot mantle upwellings may produce a low-density/low-viscosity layer. Density inversion in Archean mantle may also be caused by subduction-induced viscous underplating of depleted oceanic sublithospheric mantle (Perchuk et al., 2020). The heterogeneity of mantle upwelling or proto subduction can cause thickness variation of the CBL in the mantle.

Our model setup at time zero differs from a xenolith-constrained model for cratonic lithosphere with shallower CLAB than TLAB based on the presence of fertile peridotites in the lower portion of the conductive lithosphere (Lee et al., 2005) as commonly used in geodynamic models of cratonic stability (Cooper et al., 2004; Lenardic et al., 2003). In our interpretation, the situation with shallower CLAB than TLAB develops during cratonic evolution through thinning of the CBL and thickening of the TBL. Continuous interaction between mantle convection and the lithospheric root leads to metasomatic modification of the lower lithospheric portion (Aulbach et al., 2017; Griffin et al., 2003), thereby reducing the thickness of the CBL. At the same time, secular cooling of the mantle and radiogenic depletion of the lithospheric root increase the thickness of the TBL.

2.2. Model Geometry and Mantle Thermal Structure

The model (4,000 km × 660 km) includes a 40 km thick crust, lithospheric mantle with different thicknesses in the left domain (x = 0-2,400 km) and the right domain (x = 2,600-4,000 km) of the model, and sublithospheric mantle down to 660 km depth. At model time zero (Archean formation of a craton), the thickness of the CBL

21699356, 2023, 4

is 195 km in the thick proto-lithosphere domain and 55 km in the thin proto-lithosphere domain, with a linear change in the transition zone between the two domains. This model is an "end-member model" which demonstrates the formation of the channel flow (Figure 2, see Table S1 in Supporting Information S1 for all model parameters) due to a big difference (140 km) of CBL thickness in the model. Models with smaller differences (25–85 km) are investigated in the next sections.

Our assumption, that initially the CLAB is deeper than the TLAB, implies the presence of a partially molten material in the lower part of the Archean CBL (i.e., the layer between the TLAB and the CLAB) which may form a secondary convection system within the chemical lithosphere root with complex geotherms (Sleep, 2003a, 2003b). We simplify the geotherm shape by (a) assuming a continuous gradient of 0.3° C km⁻¹ in all convecting mantle layers, including the continental root (Sleep, 2003b) and (b) initially placing the TLAB at 90 km depth (Herzberg et al., 2010), that is maintaining CBL > TBL in both thick and thin proto-CBL domains (Figure 2c). Therefore, initially the thick domain has a 145 km vertical difference between the CLAB (at 235 km depth for a 40 km thick crust) and the TLAB, while in the thin domain this vertical difference is 5 km only. Latent heat due to partial melting in the continent root gives rise to the complexity of the temperature distribution in the continent root. The effect of latent heat is neglected in the reference model but is discussed in Section 3.2.4.

We assume that the CBL and sublithospheric mantle have different density and heat production (Figure 2b and Table S1 in Supporting Information S1): depleted cratonic lithospheric mantle with density of 3,300 kg m⁻³ is 30 kg m⁻³ (1%) less dense than sublithospheric mantle (Carlson et al., 2005), and has much higher radioactive heat production (0.2 versus 0.02 μ W/m³) than present values. The presence of a compositionally depleted, low-density cratonic lithospheric root is required to initiate the process, since the CLAB perturbation at the thick-to-thin proto-CBL transition triggers convective instabilities in the sublithospheric mantle (Section 3.1).

At model time zero the mantle potential temperature is 1,500°C (Figure 2c), that is ca. 200°C higher than at present (Nisbet et al., 1993). Due to secular cooling of the Earth and high rate of radioactive heat generation in Archean lithosphere, a 200°C lithosphere temperature anomaly converges to the same geotherm after 1 Gyr (Michaut & Jaupart, 2007). We incorporate this effect through a time-dependent heat generation model. At time

t, the rate of radioactive heat generation is $N_t = N_0 e^{\frac{-t}{\tau}}$, where N_0 is the concentration of heat producing elements at model time zero, and τ is the mean lifetime. The four major radioactive nuclides ⁴⁰K, ²³²Th, ²³⁵U, and ²³⁸U with different half-life times are present in different proportions which leads to isotopic heat production rates with one order of magnitude difference (Van Schmus, 1995). Since the total radioactive heat production in the young Earth was dominated by ⁴⁰K and ²³⁵U (see Figure S1 in Supporting Information S1), we adopt a half-life time of 1 Gyr for the radioactive nuclides. Therefore, at time zero the heat generation rate is about 4 times higher than present cratonic values (Artemieva & Mooney, 2001). The radioactive heat sources are modeled as Lagrangian particles advecting through time. Temperature at the lower boundary decays with time at a rate of 0.1°C/Myr from the initial temperature of 1,698°C at time zero (Figures 2a and 2c).

3. Results

3.1. Formation of the Channel Flow

The density decreases with depth due to increasing temperature creates density inversion in the mantle and triggers the development of convective instabilities in the sublithospheric mantle (Figure 2b). Their intensity is different below the thick and the thin proto-CBL domains, despite at time zero the model has a horizontally homogeneous temperature field. The only lateral heterogeneity in the mantle is associated with the transitional zone (x = 2,400-2,600 km) from the thick-to-thin proto-CBL domain, where the depth of the CLAB has a steep change of 140 km over a 200 km distance.

Due to its larger thickness and larger temperature difference between the top (CLAB) and the bottom (660 km depth), the sublithospheric mantle outside the thick proto-CBL domain (x > 2,600 km) has a higher Rayleigh number than below the thick proto-CBL domain (x < 2,400 km) (Figure 2c). This triggers edge convection beneath the transitional zone (x = 2,400-2,600 km) in the sublithospheric mantle (Figures 2b and 3a), which reduces the pressure in the mantle around the thick proto-CBL domain edge as compared to the pressure within the thick proto-lithosphere far from the transition zone (x = 500-2,200 km) (Figure 3a, top). As a result, the horizontal pressure gradient in the lower chemical lithosphere creates channel (Poiseuille) flow at depths between ca. 100 and 230 km (Figures 3a and 3a').





Figure 3. Snapshots of the end-member model state after 0.4 Myr (a), 3.6 Myr (b), 19.8 Myr (c), and 2487.7 Myr (d). Snapshots show the square root of second invariant strain rate. Pressure *P* profiles at depth of 90 km are plotted above each time snapshot. Gray arrows show velocity of material motion (scale shown above each snapshot); note that the scale varies with time due to mantle cooling and viscosity increase. Orange lines—temperature isotherm of 1,400°C, which closely follows the TLAB. Small circles in the upper mantle placed with a 500 km distance step illustrate mantle deformation. (a) Convective instability at the thick proto-CBL domain edge produces a low-pressure zone to the right of the thick domain. It drives the rheologically weak, high-temperature chemical lithosphere root toward the thin proto-CBL region, thus forming a continuous channel flow over a few thousand kilometers distance. The insert (white box on the left) shows a typical Poiseuille flow for stress exponent n = 1 (blue), 3 (orange), and 5 (red). A higher *n* produces a less deformed central part of the channel. (b) In the thick proto-CBL domain, the channel flow forms one strong shear zone (labeled s1) within the chemical lithosphere at a depth of ~100 km and another (labeled s2) at the top of the sublithospheric mantle. (c) The driving pressure gradient in the channel decreases in magnitude with time, reducing the intensity of the channel flow. (d) After ~2.5 Gyr of model evolution, TLAB has deepened to ~220 km depth; shear zone s1 is preserved as frozen-in strain at mid-lithosphere depths, while the sublithospheric shear zone s2 is destroyed by the underlying mantle convection.

The top of the flow marks shear zone s1 (Figure 3b), where the integrated shear strain over time (shown by small circles in Figure 3) gradually increases from the interior of the thick proto-CBL domain (left side) to its edge and the off-craton domain (right side) (Figure 3c). Another shear zone (labeled s2 in Figure 3b) develops at the base of the channel flow within the sublithospheric mantle, slightly below the CLAB.





Figure 4. Depth profiles for model evolution at x = 1,500 km (thick proto-CBL domain) for model times 0.4–19.8 Myr. Plots shows horizontal material velocity (a), strain rate (b), effective viscosity (c), and temperature (d). The top and bottom of the shear zone created by the channel flow in the cratonic root are marked by s1 and s2 (b, c). Dashed lines in (d) show dry and wet peridotite solidus with 300 ppm water content.

As the CBL material migrates rightwards to beneath the CLAB of the thin proto-CBL domain (Figures 3a–3c), deeper (y = 235 - 660 km) mantle material moves from the thick domain edge ($x = \sim 2,500$ km) leftward into the interior of the thick domain, where it rises to compensate the rightward-moving mass in the channel flow. This forms a large-scale circulation of mantle material (Figure 3a). With time, a relatively localized convection cell at time 0.4 Myr around the craton edge (Figure 3a) breaks into smaller convection cells, one propagating to the interior of the thick proto-CBL domain (leftwards) and the other to the thin proto-CBL domain (rightwards) (Figures 3b and 3c).

After 19.8 Myr, the vertically aligned circular markers at time zero (Figure 3a) in the sublithospheric mantle (depth y > 250 km) and in shallow lithosphere (y = 100 km) have an offset of 100s km with respect to the markers in the channel center (y = 150–200 km) (Figure 3c). Further, the markers at the channel top within the CBL are intensively stretched in the horizontal direction, while shear strain reduces with depth. The intensity of the edge convection wanes after 19.8 Myr of model time, when the maximum pressure gradient at the top of the channel has dropped by more than 15 MPa since model time zero. After ~2.5 Gyr of evolution, the top shear zone s1 remains preserved within the interior of the chemical lithospheric mantle, while shear zone s2 within the sublithospheric mantle disappears due to convective mixing (Figure 3d) as further discussed in Section 4.

The temporal evolution of the channel flow is further illustrated by depth profiles of material velocity, strain rate, effective viscosity, and temperature (Figures 4a-4d) at the thick proto-CBL part of the model (at distance x = 1,500 km). The top of the channel flow (s1) deepens at a rate of 1.5 km/Myr from its initial depth of 90 km depth (TLAB) at 0.4 Myr to a depth of 120 km at time = 19.8 Myr (Figures 4a-4c). While strain rate in CBL decreases by two orders of magnitude, CBL viscosity increases by one order of magnitude in ca. 20 Myr (Figure 4c). These changes are caused by a reduction of the pressure gradient in the channel (Figures 3a-3c, top panels), despite a ~40°C temperature increase in the channel (Figure 4d) which would otherwise decrease the viscosity.

3.2. Testing Parameter Space for the Channel Flow Formation

We demonstrate a broad stability of the channel flow mechanism by testing a series of model setup parameters (see Table S2 in Supporting Information S1 for overview). It is outside the scope of the present study to investigate the full parameter space for channel flow formation, since our goal is to demonstrate that this new mechanism for the MLD origin is plausible.

3.2.1. Difference Between CBL Thickness

Continent-scale channel flow forms in the lower part of Archean CBL by a horizontal pressure gradient at the craton margin caused by a downwelling mantle flow near the thick-to-thin proto-CBL transition. Mantle downwelling is triggered by a density inversion in the mantle (Figure 2b) which leads to differences in intensity of mantle flow below the thick and thin proto-CBL domains. Therefore, the initial contrast in the CLAB depth in two domains may play a critical role in the model evolution. We test model results for initial CLAB depth in the

7 of 19

thin proto-CBL domain of 210, 170, and 150 km, and 95 km depth (95 km depth in the end-member model), while keeping the CLAB of the thick proto-CBL fixed at 235 km depth as before (Figure 5).

The thermal expansion coefficient of mantle material affects both the buoyancy of the continental root and the vigor of mantle convection. Laboratory measurements on forsterite show that the thermal expansion coefficient reduces from 2.4×10^{-5} to 1.3×10^{-5} K⁻¹ for a pressure reduction from 1 GPa (top of the upper mantle) to 18 GPa (bottom of the upper mantle) (Chopelas, 1990). The value used in the end-member model $(1.0 \times 10^{-5}$ K⁻¹) better represents the deep part of the mantle. With all other parameters fixed as in the end-member model, numerical experiments with 3.0×10^{-5} K⁻¹ (Figure 5a) produce channel flow like Figure 3b. A thermal expansion coefficient of 3.0×10^{-5} K⁻¹ is used in all other the studies of CBL thickness differences across the transition zone in Figure 5.

Decreasing the CBL thickness contrast across the transition zone from 140 km (Figure 5a) to 25 km (Figure 5d) reduces the pressure difference in the channel from 20 to 5 MPa, thus reducing the mantle material flow motion rate by one order of magnitude (Figures 5a–5d). Although the mantle material motion rate is deceased due to a smaller CBL thickness contrast than that in the end-member model, a continuous channel flow at depths of ~100–200 km still develops in the CBL. After ~20 Ma, material in the s1 shear zone (the channel top) of the model with a 65 km contrast in CBL thickness across the transition zone (Figure 5c) generates >10 km dislocation at x = 1,500 km, which is about 1,000 km away from the convection center at x = 2,500 km (see circles deformation at x = 1,500 km in Figure 5c). However, in the model with a chemical lithosphere thickness of 210 km in the thin domain, there is almost no shear zone developed at x = 1,500 km (Figure 5d). If the viscosity in the sublithospheric mantle is reduced by a factor of 5, with all other parameters unchanged, the model in Figure 5d with a thickness contrast of 25 km produces a continuous the shear zone s1 at a depth of ~100 km as well due to lower viscosity and higher Rayleigh number in the sublithospheric mantle, where there is a more rigorous mantle convection (Figures 6a, 6b versus Figure 5d).

3.2.2. TLAB Depth and Density Contrast

We test how the initial TLAB depth (and therefore, initial cratonic geotherm, Figure 2c) affects the model evolution. Changing the initial TLAB depth from 90 km in the end-member model to 150 km does not prevent formation of the channel flow (Figure S3c in Supporting Information S1). However, in this case the top shear zone (s1) forms at ~150 km, that is, deeper than in the reference model. If the TLAB depth is too shallow, that means there are higher temperatures in the CBL than the end-member model, we do not observe the channel flow, but instead convection within the continent root. Whether or not the convection in the continent root interacts or merges with the sublithospheric mantle convection, depends on the density contrast between CBL and the mantle below and the Rayleigh number of the sublithospheric mantle (Jaupart et al., 2007; Sleep, 2003b) (see more discussions in next sections). If the density contrast in the end-member model is reduced to 5 kg/m³, the continent root will be destroyed, and no channel flow is observed.

3.2.3. Width of the Thick-To-Thin Proto-CBL Transition

The width of the thick-to-thin proto-CBL transition does not affect the maximum pressure drop at the craton edge (Figure 7) and therefore it does not control the conditions for the formation of a channel flow in the CBL. However, the width of the transition affects the style of mantle flow pattern. We illustrate this by a model with a wide CLAB vertical step of 145 km over 1,800 km distance (at x = 800-2,600 km) as compared to the end-member model with a narrow 145 km CLAB step over a 200 km distance at x = 2,400-2,600 km (Figure 3). A very broad transition restricts most of the deformation to the transitional zone (Figures 7a and 7b). After 6.0 Myr (Figure 7c), a large-scale convection cell near the right end of the transition (x = 2,300 km) starts splitting into two smaller cells with the opposite propagating directions, while the channel flow continues.

3.2.4. Mantle Melting

Our model does not consider melting effects, although melt extraction may consume latent heat and thereby lower temperature in the lithospheric root. We test this in two experiments (Figure S2 in Supporting Information S1) with wet and dry solidus. The compensation of latent heat in a melting system is a complex process. If partial melts and solid grains are in equilibrium, that is, the temperature in the melting system is equal to dry (or wet) solidus of olivine, the continental lithospheric root can still accommodate large-scale channel flow at a temperature lower than in the end-member model (Figure S2 in Supporting Information S1).





Figure 5. Snapshots of models with the CLAB depth of the thin proto-CBL domain of 95 km (a), 150 km (b), 170 km (c), 210 km (d) after ~20 Myr model evolution. The thick proto-CBL domain has the same chemical lithosphere thickness as the end-member model of 235 km. Notations as in Figure 3. The model has a thermal expansion coefficient of $3 \times 10^{-5} \text{ K}^{-1}$, which is triple value of that in the end-member model. Increasing thermal expansion does not affect the channel flow pattern (a). Decreasing the thickness contrast across the transition zone reduces the channel flow motion rate due to lower pressure gradient in the channel.

3.2.5. Delayed Formation of Channel Flow in the Archean

Model geometry and variations in mantle physical properties control a complex flow pattern, which may be different from those discussed above (Figures 3, 5–7 and Figures S2, S3 in Supporting Information S1). In particular, the formation of the channel flow depends on the initial conditions, which are poorly constrained for the Archean Earth. Among many tested models, we found a few cases, when a large-scale channel flow does not form at the early stages of model evolution, but develops after tens of million years. The model, named as M_den15_exp2, with the delayed channel flow formation has the same parameters as the end-member model (Tables S1 and S2





Figure 6. Snapshots of the model evolution after 3.7 Myr (a) and 28.6 Myr (b). Compared to the model in Figure 5d (Figure 3), the viscosity in the sublithospheric mantle is reduced by a factor of 5. Notations as in Figure 3. The model develops a channel flow in the chemical lithosphere root, and the flow is more rigorous than that in Figure 5d, though they have the same geometry in the initial setup.

in Supporting Information S1 and Figures 2 and 3), except that the density contrast across the CLAB is reduced from 30 to 15 kg/m³, and the thermal expansion coefficient of 2.0×10^{-5} K⁻¹ is double of that in the end-member model. Due to the higher value of thermal expansion and higher Rayleigh number (see Sections 3.2.6), the evolution of this model (Figure 8a) creates a very vigorous edge convection at *x* = 2,000–3,000 km.

The initial channel flow formed during the first 1.0 Myr is <500 km long and most of the deformation localized at the transitional part (Figures 8a and 8b). The leftward movement of this convection cell from the thick proto-CBL domain edge toward its interior deforms the shape of the CLAB. Vigorous convection below the thick proto-CBL domain (x = 700-1,200 km; Figure 8c) increases pressure in the directly overlying channel segment (Figure 8c, top). The rising pressure creates a large-scale channel flow in the chemical (depleted) lithospheric mantle to the right from the convection cell center. After 48.5 Myr of model evolution, the channel flow extends to the whole model length with a pressure contrast of 15 MPa between the maximum and minimum values across the model domains (Figure 8d).

The pressure bulge above the convection cell center in the sublithospheric mantle does not form in the end-member model (Figures 3a-3c), which instead shows a monotonic pressure decay from the interior of the thick proto-CBL domain (model left) to its edge (model right). This difference is caused by the more vigorous convection in model M_den15_exp2 (Figure 8), which causes deformation of the CLAB geometry during the leftward propagation of the sub-CLAB convecting cells. If the compositional density contrast across the CLAB is further reduced, for example, by metasomatism of depleted cratonic mantle, the cratonic root will be destroyed by mantle convection, as reported in numerous studies of craton stability and longevity (Cottrell et al., 2004; Lenardic et al., 2003).

3.2.6. Mantle Rayleigh Number, Convection Vigor, and Channel Flow

Formation of a depleted mantle may be a progressive process, which may delay the channel flow formation. Early stages of the lithospheric mantle depletion may have led to an initially small contrast in density and lithospheric mantle thickness between cratonic and noncratonic domains. These small contrasts result in a very slow channel flow (Figures 6 and 8). Importantly, an experiment with a 25 km difference in CBL thickness demonstrates that even small perturbations at the CLAB can trigger channel flows in cratonic roots during or after the proto-craton formation. While mantle heterogeneities (e.g., mantle fluids) tend to perturb the CLAB shape, the resultant channel flow in cratonic roots tends to flatten it.





Figure 7. Snapshots of model evolution after 0.4 Myr (a), 3.6 Myr (b), and 6.0 Myr (c) for a broad transitional zone at x = 800-2,600 km. Other model parameters are the same as for model evolution in Figure 3. Notations as in Figure 3. A broad chemical LAB (CLAB) ramp at the transitional zone produces larger convection cells and also leads to the characteristic channel flow in the continent root.

Model M_den15_exp2 (Figure 8) demonstrates that the early stage evolution, which is strongly controlled by the initial conditions, does not define whether or not channel flow may form at later stages. Therefore, we track changes in model key physical properties through time, especially at stages when convection cells begin to deform the CLAB geometry from below and/or above. We adopt a standard convection modeling approach to describe model evolution by dimensionless numbers, which incorporate most critical model parameters, given their large number other approaches are not justified.

The dimensionless Rayleigh number $Ra = \frac{\rho_r g \alpha \Delta T h^3}{\kappa \eta_r}$ and lithosphere Buoyancy number $B = \frac{\Delta \rho}{\rho_r \alpha \Delta T}$ characterize mantle flow patterns. Here *h* is thickness of the continent root, ρ_r its chemical density, $\Delta \rho$ the chemical density contrast across the CLAB, ΔT the maximum temperature difference within the continent root, κ thermal diffusivity, α thermal expansion coefficient, and η_r average viscosity in the cratonic root.

We calculate *Ra* and *B* for a set of models with various $\Delta \rho$, η_r , and α , preserving the same geometry as in the end-member model (Section 2.2). As ΔT and η_r change both in space and time, we limit observations to the area within the thick proto-CBL domain interior. For example, at 1.0 Myr of the M_den15_exp2 model evolution (Figure 8a), the mantle below its interior (x = 0-700 km) remains stable, while below its edge (x > 2,000 km) the mantle is involved in strong convection. Since averaging over the whole cratonic lithosphere root hides local heterogeneities, we only average the properties over ~500 km distance within the thick proto-CBL interior.

The results fall into two major convection domains, the stable domain and the whole-layer mixed convection domain (Figure 9). In the stable domain, the thick continental root is neither affected by convection from below

11 of 19





Figure 8. Snapshots of the evolution of model M_den15_exp2 after 0.4 Myr (a), 1.0 Myr (b), 11.4 Myr (c), and 48.5 Myr (d). Compared to the end-member model (Figure 2), the model has the half density contrast across the chemical LAB (CLAB) (15 kg/m³) and double value of thermal expansion coefficient of 2×10^{-5} K⁻¹ (see details in Tables S1 and S2 in Supporting Information S1). Notations as in Figure 3. At the early stage (a, b), the model does not develop a channel flow in the chemical lithosphere root; a channel flow forms at later stages (c, d) due to a pressure increase directly above the convection cell centered around $x = \sim 800-1,000$ km.

nor develops small-scale convection inside the continent root, which allow channel flow to develop. The other end-member case, with mantle convection in the layer between the CLAB and 660 km depth, includes delamination of the continent root into the hot, deep mantle (Figure 9b). The boundary between the two domains is defined by a critical Rayleigh number (Ra_c) (dashed line in Figure 9a). The Ra_c linearly increases from ~200 to ~1,000 (with some outliers) for the Buoyancy number between 0.0 and 0.4–0.5 and is almost constant ($Ra_c \sim 1,000$) for B > 0.5. The latter value is close to $Ra_c \sim 1,100$ as reported for a typical Rayleigh-Benard convection with one rigid and one free slip boundary (Pellew & Southwell, 1940) and in comprehensive theoretical instability studies (Jaupart et al., 2007). Similar value of $Ra_c \approx 1,100$ was reported in laboratory studies of convection instability for

10.1029/2022JB026202





Figure 9. (a) Rayleigh number (*Ra*) versus Buoyance number (*B*) for a set of models with variable physical parameters. See text for details. The critical Rayleigh number (dashed line) separates the stable domain (black symbols) from the whole-layer mixed convection domain (open symbols). (b) An example of whole-layer mixed convection. The model M_den05_exp3 (see Tables S1 and S2 in Supporting Information S1 for details) is the same as the end-member model except for a smaller density contrast across the chemical LAB (CLAB) (5 kg/m³) and triple value of thermal expansion coefficient ($3 \times 10^{-5} \text{ K}^{-1}$). (c) Snapshot of model M_den15_ exp2 evolution at time = 294.7 Myr, which shows a stratified oscillatory CLAB interface. Dark blue—chemical boundary layer (depleted lithosphere), light blue— sublithospheric mantle. White arrows—velocity of material motion with the scale atop the snapshots.

two-layered fluids; this critical value is almost independent on the viscosity change across the material interface (Cottrell et al., 2004), that corresponds to the CLAB in our study.

The exact position of the critical $Ra_c - B$ curve has some uncertainty (Figure 9a). Close to the critical boundary, the two mantle layers (depleted lithospheric and fertile asthenospheric) are stratified, but the CLAB interface oscillates (Figure 9c). The convective overturning instability essentially drives the dynamics of the model (Figure 2b), while viscous forces and thermal diffusion stabilizes the density contrast across the CLAB and work as a restoring force. The efficiency of heat diffusion in the depleted cratonic root affects the mantle flow pattern. Large-scale channel flow in the cratonic root is less efficient for heat diffusion than mantle convection, which leads to accumulation of extra heat in the cratonic root when radioactive heat generation rate is high (Figure 4d). It increases the Rayleigh number and causes oscillation of the CLAB interface (Figure 9c).



4. Discussion

Our numerical model provides a plausible dynamic process for the formation of MLD by channel flow within the cratonic lithosphere during the Archean. Channel flow typically forms below 80–120 km depth (deeper at later times due to cooling of the Earth and decay of the driving pressure gradient) with the bottom at depths >200 km. The top (s1) and bottom (s2) of the channel flow experience maximum shear deformation, whereas its middle part (120–200 km) is less deformed (Figures 3–8). Therefore, the main signature of channel flow is frozen-in deformation at the channel top. It will manifest itself in a tens of km thick layer by changes in for example, grain size, seismic velocities, seismic and electrical anisotropy (Figures 1a and 1b). The bottom of the channel flow is likely to be destroyed by mantle convection. It may be preserved in rare cases of an exceptionally deep depleted cratonic roots and will manifest itself similar to the top shear zone.

4.1. Channel Flow and Seismic Low-Velocity Zones

Intensive shearing at the top of the channel flow (layer s1, depth 90–120 km, Figures 3 and 7) corresponds to a low-viscosity zone (Figure 4c) at depths consistent with the depth to seismic velocity discontinuities (Rader et al., 2015; Thybo & Perchuć, 1997) (Figure 1a). Although a low viscosity does not necessarily require a low seismic velocity (Karato, 1995), the seismic low-velocity zone is generally regarded as a rheologically weak upper mantle zone (Thybo, 2006). Mineral physics experiments on melt-free polycrystalline aggregates of Fo90 olivine indicate that shear modulus decreases with reducing grain size, for example, at 1,300°C, the shear modulus is decreased by 3–4% if the grain size is reduced from 10 to 1 mm at a period of 10–100 s (Jackson & Faul, 2010). Grain size reduction my be caused by intensive shearing in the mantle (Bystricky et al., 2000), as evidenced by mantle xenoliths from the Kaapvaal craton with a drop in olivine grain size from 12 to 5 mm at 130–160 km depth (Mercier, 1980). However, at 1,100°C, the shear modulus is decreased only by ~2%, corresponding to ~1% reduction in shear-wave velocity, for the grain size reduced from 10 to 1 mm (Jackson & Faul, 2010). The secular cooling of the earth may make the seismic velocity discontinuity more difficult to be detected at present than at the Archean, if the grains size in the shear zone has not changed significantly since the formation of cratons (Speciale et al., 2020). Therefore, the grain size reduction caused by channel flow alone is difficult to explain current observations.

4.2. Channel Flow, Phase Transitions, and Fluid Migration

Volatile-rich melts percolating through the lower lithosphere may concentrate at some depths, leading to accumulations of minerals with low seismic velocities (pyroxenes, phlogopite, amphibole, carbonates) (Aulbach et al., 2017). The presence of these minerals at mid-lithosphere depths was proposed as an explanation of MLD observed in seismic *S*-RFs (Selway et al., 2015). However, our compilation shows little correlation between the depth of seismic MLD and the petrologically constrained top of the metasomatically reworked lithosphere mantle in nearly all cratons (Figure 1a). The top of the melt metasomatism is everywhere deeper than the geophysical observations of MLD (Figure 1a). Instead, seismic observations of MLDs are likely to be related to spinel-garnet transition (Figure 1a).

Both shear zones (s1 at 80–120 km and s2 at 200 km) form below low-deformed, high-viscosity lithospheric mantle. Above the channel top (s1) high viscosity is due to low temperature; above s2 it is due to high strength from low strain rate for dislocation creep in the central part of the channel (Figure 3a'). Volatile-rich melts from asthenosphere (>200 km) may be blocked by the central part of the channel flow, while at shallower depth (60–140 km) volatile fluids may be released by chemical reactions (Scambelluri et al., 2006; Skirrow et al., 2018) and can be capped by high-viscosity lithosphere above the top shear zone (s1). The top shear zone (s1) at ~100 km coincides with the depth of pargasite dehydration solidus, 1,100°C at ~90–100 km for old cratons, and the fluids rising from underlying asthenosphere may accumulate at the base of the pargasite stable zone at a depth around 100 km, preventing the rest of the mantle at a shallower depth than 100 km from rehydration and chemical alteration (Kovács et al., 2021). Both mechanisms may work together to create a barrier to rising fluids and accumulate fluids at ~100 km.

The thickness of the less deformed middle part of the channel is comparable to the thickness of intra-lithosphere anomalous layers in cratons from seismic, MT and xenolith data (Figure 1b), which varies from 20 to 70 km with a global average of \sim 50 km. Our model indicates that when conditions are suitable to produce channel flow (i.e.,



21699356, 2023, 4

sufficient pressure gradient), the depth of shear zone s1 depends on mantle temperature (Figure 3 and Figure S3c in Supporting Information S1). If the craton is later reheated, there could be multiple s1 shear zones.

4.3. Channel Flow, Seismic Anisotropy, and Contrasting Seismic Signals From MLD and LAB

Channel flow causes strong radial anisotropy at the top (s1) and base (s2) of the channel, but not in-between, where strain rate is very low as typical of Poiseuille flow (Figure 3a'). This is consistent with observations of stronger radial anisotropy in the shallow mid-lithosphere mantle than at around the LAB in the cratons in western Australia (Sun et al., 2018). Radial anisotropy may produce a seismic MLD as observed in *S*-RF (Selway et al., 2015). Dislocation creep in non-Newtonian material with the stress exponent n > 1 (Figure 3a') would enhance a localized shear zone s1 and associated seismic anisotropy (Karato, 1995).

Our channel flow model predicts two distinct motion patterns for mantle material in the young and present Earth (Figures 3a–3c and 9). In the young Earth, a continuous channel flow forms beneath the TLAB at a depth of ~100 km, while after billions of years of mantle cooling, multiple small convection cells underlie the TLAB of present earth at a depth of ~200 km (Figures 3d and 9). Therefore, in the Archean a nearly horizontal shear below the TLAB due to the channel, while the mantle flow below the present TLAB is characterized by alternating upwelling and downwelling (Figure 3d). Higher horizontal strain and less vertical shear developed in the channel top (s1) than the channel bottom (s2) should produce a stronger radial anisotropy ($V_{SH} > V_{SV}$) in the top than the bottom.

4.4. Implications for Mantle Dynamics

Channel flow cannot form in modern cratonic settings: cold and rigid cratons with high-viscosity lithosphere roots tend to resist the development of a channel flow, even in cases of significant horizontal pressure gradient, for example, from edge convection (King & Ritsema, 2000). We note that during the evolution of the end-member model, at ca. 20 Myr, geotherms develop an intriguing curvature at ~200 km depth (Figure 4d), when temperature in the CBL is higher than in sublithospheric mantle. Such kinked, hot lithospheric geotherm may exist during a period of very high rate of radioactive heat generation in the Precambrian lithosphere (Michaut & Jaupart, 2007). High temperature in the cratonic root reduces its viscosity and facilitates formation of channel flow once a sufficient pressure gradient develops.

In our numerical model, we assume that this thermal condition is satisfied in the Archean with mantle temperatures ca. 200°C higher than at present. However, large-scale mantle thermal anomalies may introduce additional heat at the base of cratons which may increase temperature at the lithosphere base by 100–150°C, possibly over distances of 500 km away from the impact center (Sleep, 2003a). If the increased temperature at the continental lithospheric root reaches the wet solidus, channel flow may be triggered in the lower cratonic lithosphere too (Figure S2 in Supporting Information S1). This may contribute to explaining some enigmatic features, such as high (2–7%) shear-wave velocity anomalies which appear to continue from the ancient continental cratonic roots of the southern African cratons to beneath the oceanic basins 500-1,300 km away from the cratonic margin (Grand, 2002). If the mantle thermal anomaly associated with the separation of the South American and African continents increased the temperature of the cratonic root by 100–150°C, it would have reduced the mantle viscosity by approximately two orders of magnitude. Mantle edge convection tends to create a low-pressure zone in the upper mantle at cratonic margins with respect to cratonic interior, which may cause a weakened cratonic root to flow toward the ocean like the pattern observed in the reference model (Figures 3a-3c). If a continental plate moves in the direction opposite to the channel flow, it may enhance shearing at depths of the top of the channel flow (s1), and this process may explain observations of high seismic velocity anomalies such as cratonic continent roots beneath the oceans around southern Africa (Wang et al., 2017).

As reviewed in Section 1, seismic, MT, and xenolith data report variable depths to the top of the intra-lithosphere discontinuities in cratons (Figure 1). Our numerical experiments generate the top shear zone (s1) in a relatively narrow depth range (100–130 km), which fits observations of seismic anisotropy and the seismic low-velocity zone from controlled-source seismic experiments better than other intra-lithospheric geophysical features. Our model provides a generic framework for understanding the origin of seismic radial anisotropy; it also provides a physical mechanism to explain the major MLD features observed from *S*-RF (Selway et al., 2015). It is possible that, during later cratonic evolution, various local tectonic events can significantly modify preexisting frozen-in





Figure 10. Sketch (not to scale) illustrating mantle evolution through time from a newborn thick cratonic lithosphere to the present. It shows how the footprints of the once existing channel flow structure (black dots) remain preserved in the present mid-lithospheric mantle. Secular cooling of the Earth's interior leads to a change in the upper mantle convection style and its interaction with the cratonic chemical boundary layer (lithospheric root). LAB—lithosphere-asthenosphere boundary, TLAB—thermal LAB, CLAB—chemical LAB.

shear zones. For example, a high-resolution teleseismic *S* wave study imaged two MLDs around the Midcontinent Rift and the southern Appalachians, that is, in the regions that have experienced significant tectonomagmatic modifications since cratonization (Liu & Shearer, 2021). Phanerozoic rifting processes in cratons can raise temperature in the continental root and create a CBL thickness contrast, thus supporting the formation of channel flow in the continent root as well (Huismans & Beaumont, 2011). The reheating event may cause deeper and smaller scale shear zones than the original channel flow and produce multi MLDs, but generally over much shorter distances than the originally formed MLD.

5. Conclusions

We propose that the MLD was formed by channel flow in the Archean Earth and test this hypothesis by two-dimensional thermomechanical numerical modeling. The critical aspects of our model are that at time zero, the thermal LAB (TLAB) is shallower than the chemical LAB (CLAB), and the CLAB transition between thick and thin proto-lithosphere triggers mantle flow in the lithospheric (above the CLAB) and sublithospheric mantle (Figure 10).

- 1. High mantle temperature in the Archean Earth is critical for the formation of the channel flow in a thick buoyant cratonic lithosphere. Continent-scale channel flow forms in the lower chemical lithosphere due to a horizontal pressure gradient originating from mantle convective instabilities around the craton edge.
- 2. The channel flow has a Poiseuille-type deformation style and creates deformation zones at 80–120 km and 200–250 km depths. The top deformation zone freezes-in at mid-lithosphere depths, while the bottom zone may be destroyed by mantle convection. The less deformed central part of the channel may become a barrier to fluids rising from the sublithospheric mantle.
- 3. The mid-lithosphere channel flow predicts and explains the presence of seismic low-velocity zones and seismic MLD in the cratons. Intensive shearing near the top of the channel flow may reduce olivine grain size, promote accumulation of volatile-rich melts and create frozen-in anisotropy to develop a characteristic seismic discontinuity.
- 4. The interplay between the mantle Rayleigh number and the lithosphere chemical buoyancy controls the formation of either a mid-lithosphere channel flow or mantle convection cells with possible destruction of cratonic roots.

21699356,

2023, 4, Downle

doi/10.1029/2022JB026202

by HGF GEOMAR

5. Secular cooling affects the flow structure beneath the TLAB, which switches from the early mode with a channel flow to the formation of small multiple convection cells at later stages. These multiple convection cells with alternating upwelling and downwelling may reduce the strength of radial anisotropy caused by large-scale channel flow in shallow proto-cratonic lithosphere.

Data Availability Statement

This study is based on published data, which are available as referenced in the article. The numerical open-source code Underworld2 used for the calculations is available at https://doi.org/10.5281/zenodo.1436039.

References

- Abramovitz, T., Thybo, H., & Perchuc, E. (2002). Tomographic inversion of seismic P- and S-wave velocities from the Baltic Shield based on FENNOLORA data. *Tectonophysics*, 358(1–4), 151–174. https://doi.org/10.1016/s0040-1951(02)00422-5
- Artemieva, I. M. (2009). The continental lithosphere: Reconciling thermal, seismic, and petrologic data. *Lithos*, 109(1–2), 23–46. https://doi. org/10.1016/j.lithos.2008.09.015
- Artemieva, I. M., & 'Mooney, W. D. (2001). Thermal thickness and evolution of Precambrian lithosphere: A global study. Journal of Geophysical Research: Solid Earth, 106(B8), 16387–16414. https://doi.org/10.1029/2000JB900439
- Aulbach, S., Massuyeau, M., & Gaillard, F. (2017). Origins of cratonic mantle discontinuities: A view from petrology, geochemistry and thermodynamic models. *Lithos*, 268, 364–382. https://doi.org/10.1016/j.lithos.2016.11.004
- Bystricky, M., Kunze, K., Burlini, L., & Burg, J.-P. (2000). High shear strain of olivine aggregates: Rheological and seismic consequences. Science, 290(5496), 1564–1567. https://doi.org/10.1126/science.290.5496.1564
- Calo, M., Bodin, T., & Romanowicz, B. (2016). Layered structure in the upper mantle across North America from joint inversion of long and short period seismic data. *Earth and Planetary Science Letters*, 449, 164–175. https://doi.org/10.1016/j.epsl.2016.05.054
- Carlson, R. W., Pearson, D. G., & James, D. E. (2005). Physical, chemical, and chronological characteristics of continental mantle. *Reviews of Geophysics*, 43, RG1001. https://doi.org/10.1029/2004RG000156
- Chen, C., Gilbert, H., Fischer, K. M., Andronicos, C. L., Pavlis, G. L., Hamburger, M. W., et al. (2018). Lithospheric discontinuities beneath the US Midcontinent–signatures of Proterozoic terrane accretion and failed rifting. *Earth and Planetary Science Letters*, 481, 223–235. https://doi.org/10.1016/j.epsl.2017.10.033
- Chopelas, A. (1990). Thermal properties of forsterite at mantle pressures derived from vibrational spectroscopy. *Physics and Chemistry of Minerals*, 17(2), 149–156. https://doi.org/10.1007/bf00199666
- Cooper, C. M., Lenardic, A., & Moresi, L. (2004). The thermal structure of stable continental lithosphere within a dynamic mantle. *Earth and Planetary Science Letters*, 222(3–4), 807–817. https://doi.org/10.1016/j.epsl.2004.04.008
- Cottrell, E., Jaupart, C., & Molnar, P. (2004). Marginal stability of thick continental lithosphere. *Geophysical Research Letters*, 31, L18612. https://doi.org/10.1029/2004GL020332
- Egorkin, A. V., Zuganov, S. K., Pavlenkova, N. A., & Chernyshev, N. M. (1987). Results of lithospheric studies from long-range profiles in Siberia. *Tectonophysics*, 140(1), 29–47. https://doi.org/10.1016/0040-1951(87)90138-7
- Fischer, K. M., Ford, H. A., Abt, D. L., & Rychert, C. A. (2010). The lithosphere-asthenosphere boundary. Annual Review of Earth and Planetary Sciences, 38(1), 551–575. https://doi.org/10.1146/annurev-earth-040809-152438
- Grand, S. P. (2002). Mantle shear-wave tomography and the fate of subducted slabs. *Philosophical Transactions of the Royal Society of London* A: Mathematical, Physical and Engineering Sciences, 360(1800), 2475–2491. https://doi.org/10.1098/rsta.2002.1077
- Griffin, W. L., O'Reilly, S. Y., Natapov, L. M., & Ryan, C. G. (2003). The evolution of lithospheric mantle beneath the Kalahari Craton and its margins. *Lithos*, 71(2–4), 215–241. https://doi.org/10.1016/j.lithos.2003.07.006
- Gung, Y., Panning, M., & Romanowicz, B. (2003). Global anisotropy and the thickness of continents. *Nature*, 422(6933), 707–711. https://doi.org/10.1038/nature01559
- Herzberg, C., Condie, K., & Korenaga, J. (2010). Thermal history of the Earth and its petrological expression. Earth and Planetary Science Letters, 292(1–2), 79–88. https://doi.org/10.1016/j.epsl.2010.01.022
- Huismans, R., & Beaumont, C. (2011). Depth-dependent extension, two-stage breakup and cratonic underplating at rifted margins. *Nature*, 473(7345), 74–78. https://doi.org/10.1038/nature09988
- Jackson, I., & Faul, U. H. (2010). Grainsize-sensitive viscoelastic relaxation in olivine: Towards a robust laboratory-based model for seismological application. *Physics of the Earth and Planetary Interiors*, 183(1–2), 151–163. https://doi.org/10.1016/j.pepi.2010.09.005
- Jaupart, C., Molnar, P., & Cottrell, E. (2007). Instability of a chemically dense layer heated from below and overlain by a deep less viscous fluid. Journal of Fluid Mechanics, 572, 433–469. https://doi.org/10.1017/s0022112006003521
- Jones, A. G., Lezaeta, P., Ferguson, I. J., Chave, A. D., Evans, R. L., Garcia, X., & Spratt, J. (2003). The electrical structure of the Slave craton. Lithos, 71(2–4), 505–527. https://doi.org/10.1016/j.lithos.2003.08.001
- Karato, S. (1995). Effects of water on seismic-wave velocities in the upper-mantle. *Proceedings of the Japan Academy Series B: Physical and Biological Sciences*, 71(2), 61–66. https://doi.org/10.2183/pjab.71.61
- Karato, S. I., Olugboji, T., & Park, J. (2015). Mechanisms and geologic significance of the mid-lithosphere discontinuity in the continents. *Nature Geoscience*, 8(7), 509–514. https://doi.org/10.1038/ngeo2462
- King, S. D., & Ritsema, J. (2000). African hot Spot Volcanism: Small-scale convection in the upper mantle beneath cratons. *Science*, 290(5494), 1137–1140. https://doi.org/10.1126/science.290.5494.1137
- Koptev, A. I., & Ershov, A. V. (2010). The role of the gravitational potential of the lithosphere in the formation of a global stress field. *Izvestiya Physics of the Solid Earth*, 46(12), 1080–1094. https://doi.org/10.1134/s1069351310120050
- Kopylova, M. G., & Caro, G. (2004). Mantle xenoliths from the Southeastern Slave craton: Evidence for chemical zonation in a thick, cold lithosphere. *Journal of Petrology*, 45(5), 1045–1067. https://doi.org/10.1093/petrology/egh003
- Kopylova, M. G., & Russell, J. K. (2000). Chemical stratification of cratonic lithosphere: Constraints from the northern Slave craton, Canada. *Earth and Planetary Science Letters*, 181(1–2), 71–87. https://doi.org/10.1016/s0012-821x(00)00187-4
- Kovács, I. J., Liptai, N., Koptev, A., Cloetingh, S., Lange, T. P., Matenco, L., et al. (2021). The 'pargasosphere' hypothesis: Looking at global plate tectonics from a new perspective. *Global and Planetary Change*, 204, 103547. https://doi.org/10.1016/j.gloplacha.2021.103547

YANG ET AL.

IMA and HT acknowledge the Grant from National Science Foundation of China (92055210) and partial support by the MOST special funds of China for GPMR State Key Laboratory (GPMR2019010). HY thanks the encouragement and technical support from Zurab Chemia when working on this project in Denmark, which was supported by the Large Grants from the Danish Research Council DFF-1323-00053 to IMA and FNU16-059776-15 to HT. The authors acknowledge the use of resources from the National Computational Infrastructure in Canberra. Open access publishing facilitated by Australian National University, as part of the Wiley - Australian National University agreement via the Council of Australian University Librarians.

- Lee, C. T. A., Lenardic, A., Cooper, C. M., Niu, F. L., & Levander, A. (2005). The role of chemical boundary layers in regulating the thickness of continental and oceanic thermal boundary layers. *Earth and Planetary Science Letters*, 230(3–4), 379–395. https://doi.org/10.1016/j. epsl.2004.11.019
- Lenardic, A., Moresi, L. N., & Mühlhaus, H. (2003). Longevity and stability of cratonic lithosphere: Insights from numerical simulations of coupled mantle convection and continental tectonics. *Journal of Geophysical Research*, 108(B6), 2303. https://doi.org/10.1029/2002JB001859
- Liu, T., & Shearer, P. M. (2021). Complicated lithospheric structure beneath the Contiguous US revealed by teleseismic S-reflections. Journal of Geophysical Research: Solid Earth, 126, e2020JB021624. https://doi.org/10.1029/2020JB021624
- Mareschal, M., Kellett, R. L., Kurtz, R. D., Ludden, J. N., Ji, S., & Bailey, R. C. (1995). Archaean cratonic roots, mantle shear zones and deep electrical anisotropy. *Nature*, 375(6527), 134–137. https://doi.org/10.1038/375134a0
- Mercier, J. C. C. (1980). Magnitude of the continental lithospheric stresses inferred from rheomorphic petrology. *Journal of Geophysical Research*, 85(Nb11), 6293–6303. https://doi.org/10.1029/JB085iB11p06293
- Michaut, C., & Jaupart, C. (2007). Secular cooling and thermal structure of continental lithosphere. *Earth and Planetary Science Letters*, 257(1–2), 83–96. https://doi.org/10.1016/j.epsl.2007.02.019
- Moresi, L., Quenette, S., Lemiale, V., Meriaux, C., Appelbe, B., & Muhlhaus, H. B. (2007). Computational approaches to studying non-linear dynamics of the crust and mantle. *Physics of the Earth and Planetary Interiors*, 163(1–4), 69–82. https://doi.org/10.1016/j.pepi.2007.06.009
- Nisbet, E. G., Cheadle, M. J., Arndt, N. T., & Bickle, M. J. (1993). Constraining the potential temperature of the Archaean mantle: A review of the evidence from komatiites. *Lithos*, 30(3–4), 291–307. https://doi.org/10.1016/0024-4937(93)90042-b
- O'Reilly, S. Y., Griffin, W. L., Poudjom Djomani, Y. H., & Morgan, P. (2001). Are lithospheres forever. Tracking changes in subcontinental lithospheric mantle through time. *Geological Society of America Today*, 11(4), 4–10. https://doi.org/10.1130/1052-5173(2001)011<0004:ALFT CI>2.0.CO;2
- Pellew, A., & Southwell, R. V. (1940). On maintained convective motion in a fluid heated from below. Proceedings of the Royal Society of London Series A: Mathematical and Physical Sciences, 176(A966), 0312–0343.
- Perchuc, E., & Thybo, H. (1996). A new model of upper mantle P-wave velocity below the Baltic Shield: Indication of partial melt in the 95 to 160 km depth range. *Tectonophysics*, 253(3–4), 227–245. https://doi.org/10.1016/0040-1951(95)00057-7
- Perchuk, A. L., Gerya, T. V., Zakharov, V. S., & Griffin, W. L. (2020). Building cratonic keels in Precambrian plate tectonics. *Nature*, 586(7829), 395–401. https://doi.org/10.1038/s41586-020-2806-7
- Rader, E., Emry, E., Schmerr, N., Frost, D., Cheng, C., Menard, J., et al. (2015). Characterization and petrological constraints of the Midlithospheric discontinuity. *Geochemistry, Geophysics, Geosystems, 16*, 3484–3504. https://doi.org/10.1002/2015GC005943
- Rychert, C. A., & Shearer, P. M. (2009). A global view of the lithosphere-asthenosphere boundary. *Science*, 324(5926), 495–498. https://doi. org/10.1126/science.1169754
- Savage, B., & Silver, P. G. (2008). Evidence for a compositional boundary within the lithospheric mantle beneath the Kalahari craton from S receiver functions. *Earth and Planetary Science Letters*, 272(3–4), 600–609. https://doi.org/10.1016/j.epsl.2008.05.026
- Scambelluri, M., Hermann, J., Morten, L., & Rampone, E. (2006). Melt-versus fluid-induced metasomatism in spinel to garnet wedge peridotites (Ulten zone, Eastern Italian Alps): Clues from trace element and Li abundances. *Contributions to Mineralogy and Petrology*, 151(4), 372–394. https://doi.org/10.1007/s00410-006-0064-9
- Selway, K., Ford, H., & Kelemen, P. (2015). The seismic mid-lithosphere discontinuity. Earth and Planetary Science Letters, 414, 45–57. https:// doi.org/10.1016/j.epsl.2014.12.029
- Skirrow, R. G., Van Der Wielen, S. E., Champion, D. C., Czarnota, K., & Thiel, S. (2018). Lithospheric architecture and mantle metasomatism linked to iron oxide Cu-Au ore formation: Multidisciplinary evidence from the Olympic Dam region, South Australia. *Geochemistry, Geophys*ics, Geosystems, 19(8), 2673–2705. https://doi.org/10.1029/2018gc007561
- Sleep, N. H. (2003a). Geodynamic implications of xenolith geotherms. *Geochemistry, Geophysics, Geosystems*, 4(9), 1079. https://doi.org/10.1029/2003GC000511
- Sleep, N. H. (2003b). Survival of Archean cratonal lithosphere. Journal of Geophysical Research: Solid Earth, 108(B6), 2302. https://doi.org/10.1029/2001JB000169
- Speciale, P. A., Behr, W. M., Hirth, G., & Tokle, L. (2020). Rates of olivine grain growth during dynamic recrystallization and postdeformation annealing. *Journal of Geophysical Research: Solid Earth*, 125, e2020JB020415. https://doi.org/10.1029/2020JB020415
- Sun, W., Fu, L. Y., Saygin, E., & Zhao, L. (2018). Insights into layering in the cratonic lithosphere beneath Western Australia. Journal of Geophysical Research: Solid Earth, 123, 1405–1418. https://doi.org/10.1002/2017JB014904
- Thybo, H. (2006). The heterogeneous upper mantle low velocity zone. *Tectonophysics*, 416(1-4), 53-79. https://doi.org/10.1016/j. tecto.2005.11.021
- Thybo, H., & Perchuć, E. (1997). The seismic 8° discontinuity and partial melting in continental mantle. *Science*, 275(5306), 1626–1629. https://doi.org/10.1126/science.275.5306.1626
- Van Schmus, W. R. (1995). Natural radioactivity of the crust and mantle, global earth physics: A handbook of physical constants (pp. 283–291). American Geophysical Union.
- Wang, Z., Kusky, T. M., & Capitanio, F. A. (2017). Ancient continental lithosphere dislocated beneath ocean basins along the mid-lithosphere discontinuity: A hypothesis. *Geophysical Research Letters*, 44, 9253–9260. https://doi.org/10.1002/2017GL074686

Yuan, H., & Romanowicz, B. (2010). Lithospheric layering in the North American craton. *Nature*, 466(7310), 1063–1068. https://doi.org/10.1038/ nature09332

References From the Supporting Information

- Abt, D. L., Fischer, K. M., French, S. W., Ford, H. A., Yuan, H., & Romanowicz, B. (2010). North American lithospheric discontinuity structure imaged by Ps and Sp receiver functions. *Journal of Geophysical Research*, 115, B09301. https://doi.org/10.1029/2009JB006914
- Bruneton, M., Pedersen, H. A., Vacher, P., Kukkonen, I. T., Arndt, N. T., Funke, S., et al. (2004). Layered lithospheric mantle in the central Baltic Shield from surface waves and xenolith analysis. *Earth and Planetary Science Letters*, 226(1–2), 41–52. https://doi.org/10.1016/j. epsl.2004.07.034
- Clauser, C., & Huenges, E. (1995). Thermal conductivity of rocks and minerals. In T. J. Ahrens (Ed.), Rock physics & phase relations: A handbook of physical constants (pp. 105–126). American Geophysical Union
- Debayle, E., Kennett, B., & Priestley, K. (2005). Global azimuthal seismic anisotropy and the unique plate-motion deformation of Australia. *Nature*, 433(7025), 509–512. https://doi.org/10.1038/nature03247

- Fuchs, K., & Vinnik, L. P. (1982). Investigation of the subcrustal lithosphere and asthenosphere by controlled source seismic experiments on long range profiles. In G. Pálmason (Ed.). Continental and oceanic riffs (Vol. 8, pp. 81–89). AGU Geodyn. Ser.
- Griffin, W. L., Doyle, B. J., Ryan, C. G., Pearson, N. J., O'Reilly, S. Y., Davies, R., et al. (1999). Layered mantle lithosphere in the Lac de Gras area, Slave Craton: Composition, structure and origin. *Journal of Petrology*, 40(5), 705–727. https://doi.org/10.1093/petroj/40.5.705
- Griffin, W. L., O'Reilly, S. Y., Doyle, B. J., Pearson, N. J., Coopersmith, H., Kivi, K., et al. (2004). Lithosphere mapping beneath the north American plate. *Lithos*, 77(1–4), 873–922. https://doi.org/10.1016/j.lithos.2004.03.034
- Griffin, W. L., Ryan, C. G., Kaminsky, F. V., O'Reilly, S. Y., Natapov, L. M., Win, T. T., et al. (1999). The Siberian lithosphere traverse: Mantle terranes and the assembly of the Siberian craton. *Tectonophysics*, 310(1–4), 1–35. https://doi.org/10.1016/s0040-1951(99)00156-0
- Hansen, S. E., Nyblade, A. A., Julià, J., Dirks, P. H. G. M., & Durrheim, R. J. (2009). Upper-mantle low-velocity zone structure beneath the Kaapvaal craton fromS-wave receiver functions. *Geophysical Journal International*, 178(2), 1021–1027. https://doi. org/10.1111/j.1365-246x.2009.04178.x
- Hirth, G., & Kohlstedt, D. (2003). Rheology of the upper mantle and the mantle wedge: A view from the experimentalists, inside the subduction factory (pp. 83–105). American Geophysical Union.
- Lehtonen, M. L., O'Brien, H. E., Peltonen, P., Johanson, B. S., & Pakkanen, L. K. (2004). Layered mantle at the Karelian craton margin: P-T of mantle xenocrysts and xenoliths from the Kaavi–Kuopio kimberlites, Finland. *Lithos*, 77(1–4), 593–608. https://doi.org/10.1016/j. lithos.2004.04.026
- Olsson, S., Roberts, R. G., & Böðvarsson, R. (2007). Analysis of waves converted from S to P in the upper mantle beneath the Baltic Shield. *Earth and Planetary Science Letters*, 257(1–2), 37–46. https://doi.org/10.1016/j.epsl.2007.02.017
- Ranalli, G. (1995). Rheology of the earth. Springer Science & Business Media.
- Silvennoinen, H., Kozlovskaya, E., & Kissling, E. (2016). POLENET/LAPNET teleseismic P wave travel time tomography model of the upper mantle beneath northern Fennoscandia. Solid Earth, 7(2), 425–439. https://doi.org/10.5194/se-7-425-2016
- Snyder, D. B. (2008). Stacked uppermost mantle layers within the Slave craton of NW Canada as defined by anisotropic seismic discontinuities. *Tectonics*, 27(4), TC4006. https://doi.org/10.1029/2007TC002132
- SodoudiYuan, F. X., Kind, R., Lebedev, S., Adam, J. M. C., Kastle, E., & Tilmann, F. (2013). Seismic evidence for stratification in composition and anisotropic fabric within the thick lithosphere of Kalahari Craton. *Geochemistry, Geophysics, Geosystems*, 14, 5393–5412. https://doi. org/10.1002/2013GC004955
- Vinnik, L., Oreshin, S., Makeyeva, L., Peregoudov, D., Kozlovskaya, E., Kozlovskaya, E., et al. (2014). Anisotropic lithosphere under the Fennoscandian shield from P receiver functions and SKS waveforms of the POLENET/LAPNET array. *Tectonophysics*, 628, 45–54. https://doi. org/10.1016/j.tecto.2014.04.024
- Wittlinger, G., & Farra, V. (2007). Converted waves reveal a thick and layered tectosphere beneath the Kalahari supercraton. *Earth and Planetary Science Letters*, 254(3–4), 404–415. https://doi.org/10.1016/j.epsl.2006.11.048
- Yuan, H., Romanowicz, B., Fischer, K. M., & Abt, D. (2011). 3-D shear wave radially and azimuthally anisotropic velocity model of the North American upper mantle. *Geophysical Journal International*, 184(3), 1237–1260. https://doi.org/10.1111/j.1365-246x.2010.04901.x